

September 3rd, 2021

Dear editor,

Please find attached files of revised manuscript in word format

Title: Effects of Minimally Invasive Surgery versus Laparotomy in Patients with Colon Cancer Residing in High-altitude Areas

Author: Duo Ji Suo Lang, Yang Zhen Ci Ren, Zha Xi Bian Ba

Name of Journal: World Journal of Clinical Cases

Manuscript NO: 70044

First of all, thank you for your careful guidance of this article. Revision has been made according to the suggestions of the reviewer:

Reviewer: 06110579

In recent years, due to the increasing incidence of colon cancer, the safe and effective treatment have become a research hotspot. Surgery is an important measure in the current clinical treatment of colon cancer, and laparotomy and laparoscopic minimally invasive surgery are commonly used. With improvements in minimally invasive technology and the popularization of this concept, laparoscopic surgery has been applied as an important clinical minimally invasive surgery in colon cancer. Laparoscopic surgery can reduce surgical trauma and shorten recovery time, which play an important role in the treatment. In this study, the authors examined the effect of minimally invasive surgery and laparotomy in patients with colon cancer residing in high-altitude areas. The reviewer thinks that the study is very interesting. Comments: 1. The results section in the abstract is too long. Authors should only keep some most important data, and short this paragraph. 2. Some minor language polishing should be revised. 3. Please check and edit the tables.

Thank you for your advice.

After receiving the comments, we read this article carefully, found some small loopholes in the language of the article, and shorted the paragraph of results section in the abstract. I modified the table format.

According to the suggestion, we changed the relevant description.

Thank you again for publishing our manuscript in the World Journal of Clinical

Cases.

Sincerely Yours,
Duo Ji Suo Lang